Author’s response to reviews

Title: A case report of an unusual non-mucinous papillary variant of CPAM type 1 with KRAS mutations

Authors:

Timco Koopman (t.koopman@umcg.nl)
Bart Rottier (b.l.rottier@umcg.nl)
Arja ter Elst (a.ter.elst@umcg.nl)
Wim Timens (w.timens@umcg.nl)

Version: 1 Date: 18 Sep 2019

Author’s response to reviews:

Assistant Editor's comments:

1. Consent for publication
Please obtain and provide the written consent for publication from the patient now as we cannot move forward to peer review without written consent for case reports.
Author response: a written consent by email (in Dutch, key sentence translated and signed by the corresponding author) is provided as a separate attachment (as supplementary material). The BioMed Central Consent Form has been sent to the parents, but due to holidays they will not be able to sign this form before the 22nd of September, which is the deadline of this revision. The parents will print, sign and send us the form when they return from their holiday, after which we can send you the completed form as soon as possible.

2. Authors' Contributions
Please represent authors' names using their full initials, not their full name, in the Authors’ Contributions section. If there are any duplicated initials, please differentiate them to make it clear that the initials refer to separate authors.
Author response: we have represented the authors' names with full initials and surnames in the Authors' Contributions section. There are no duplicated initials.

Please include a statement in the Authors' contributions section to the effect that all authors have read and approved the final manuscript, and ensure that this is the case.
Author response: the statement (which is ensured) has been added to the Authors' contributions section.